Author’s response to reviews

Title: Delayed-onset heparin-induced thrombocytopenia presenting with multiple arteriovenous thromboses: Case report

Authors:

Abbas Salehiomran (abbasomran2006@yahoo.com)
Abbassali Karimi (akarimi@sina.tums.AC.IR)
Hossein Ahmadi (dr.ahmadi2006@yahoo.com)
Parin Yazdanifard (parinyazdanifard@yahoo.com)

Version: 5 Date: 1 September 2007

Author’s response to reviews:

September 1, 2007
Dear Editor in chief
The journal of medical case report

I would like to thank you for reviewing my article and for your helpful comments. All of the changes have been made as your suggestion and are listed below. Also the changes are found in my revised manuscript as red font.

1- The "arteriovenous thromboses" was corrected in the title of article.
2- In the second paragraph, the word "cerebrovascular accident" was omitted and only "stroke" was used.
3- In the third sentences of forth paragraph of discussion section, the sentences was corrected as follow: Warfarin is recommended to start simultaneously with hirudin or other alternative anticoagulants except heparin. Warfarin is not contraindicated in treatment of HITT but because of its potential to make limb gangrene besides of heparin, it is recommended to administer after other anticoagulant in HITT [7, 8].
4- All of references was corrected.

With the best regards
A. SalehiOmran